Author's response to reviews

Title: Comparison of two self-reported measures of physical work load in hospital personnel: a cross-sectional study

Authors:

Kirsten Nabe-Nielsen (knn@arbejdsmiljoforskning.dk)
Nils Fallentin (nif@arbejdsmiljoforskning.dk)
Karl B Christensen (kbc@arbejdsmiljoforskning.dk)
Jette Nygaard Jensen (jnij@arbejdsmiljoforskning.dk)
Finn Diderichsen (F.Diderichsen@socmed.ku.dk)

Version: 4 Date: 28 January 2008

Author's response to reviews: see over
Dear Referees and Editorial Board, BioMed Central

Thank you for your instructive comments on my paper.

I have now revised my research article “Comparison of two self-reported measures of physical work load in hospital personnel: a cross-sectional study” based on the comment received the 18th of January 2008.

The following changes have been made:

1. Spelling error (extend → extent)
2. Additional information on the correlations between the measures of physical and psychosocial load as well as the correlations between the measures of psychosocial has been provided.

Sincerely,

Kirsten Nabe-Nielsen